Author's response to reviews

**Title:** Relationship between peripheral airway function and patient-reported outcomes in COPD: a cross-sectional study

**Authors:**

Akane Haruna (akaneh@kuhp.kyoto-u.ac.jp)
Toru Oga (ogat@df7.so-net.ne.jp)
Shigeo Muro (smuro@kuhp.kyoto-u.ac.jp)
Tadashi Ohara (tohara@kuhp.kyoto-u.ac.jp)
Susumu Sato (ssato@kuhp.kyoto-u.ac.jp)
Satoshi Marumo (marumo@kuhp.kyoto-u.ac.jp)
Daisuke Kinose (dkinose@kuhp.kyoto-u.ac.jp)
Kunihiko Terada (terada@kuhp.kyoto-u.ac.jp)
Michiyoshi Nishioka (mnishiok@kuhp.kyoto-u.ac.jp)
Emiko Ogawa (eogawa@kuhp.kyoto-u.ac.jp)
Yuma Hoshino (yuma@kuhp.kyoto-u.ac.jp)
Toyohiro Hirai (t_hirai@kuhp.kyoto-u.ac.jp)
Kazuo Chin (chink@kuhp.kyoto-u.ac.jp)
Michiaki Mishima (mishima@kuhp.kyoto-u.ac.jp)

**Version:** 7  **Date:** 4 March 2010

**Author's response to reviews:** see over
Dear Dr. Norton,

Editor,

BMC Pulmonary Medicine

Thank you for your letter dated March 4, 2010, for formatting changes required (MS number: 1958282827289294). We have carefully checked the manuscript and changed one word.

Herewith we submit our revised manuscript.

Sincerely yours,

Toru Oga, MD, PhD